Author's response to reviews

Title: Whole exome sequencing in an Indian family links cerebroretinal microangiopathy with calcifications and cysts (CRMCC) and dextrocardia with a homozygous novel CTC1 and a rare HES7 variation

Authors:

Manjunath Netravathi (sundernetra@yahoo.co.in)
Saketh Kapoor (kapoor.saketh@gmail.com)
Renu Kumari (renu.kumari@igib.in)
Pushkar Dakle (pushkar.dakle@igib.in)
Manish K Dwivedi (mk.dwivedi@igib.in)
Sumitabho Deb Roy (sumitabhodebroy@gmail.com)
Paritosh Pandey (paritosh2000@gmail.com)
Jitender Saini (jsaini76@gmail.com)
Anil Ramakrishna (anilramakrishna78@yahoo.co.in)
Devaraddi Navalli (devaraddi.navalli@rediffmail.com)
Parthasarathy Satishchandra (drpsatishchandra@yahoo.com)
Pramod Kumar Pal (pal.pramod@rediffmail.com)
Mohammed Faruq (faruq.mohd@igib.in)
Arun Kumar (karun@mrdg.iisc.ernet.in)

Version: 2 Date: 18 July 2014

Author's response to reviews: see over
To

The Editor,

BMC Medical Genetics

Subject: revised submission of our manuscript MS: 171256222136259, Research article

We are submitting our revised manuscript entitled “Whole exome sequencing in an Indian family links cerebroretinal microangiopathy with calcifications and cysts(CRMCC) and dextrocardia with a homozygous novel CTC1 and a rare HES7 variation by Manjunath Netravathi et al. for consideration as a “Research article” in BMC Medical Genetics..

Following are the point by point descriptions of the changes made

1. We have provided the email addresses of all the authors on the title page.
2. We have added Ethics statement at the end of the main text which reads as
   “Ethics statement: This research followed the tenets of the Declaration of Helsinki and the guidelines of the Indian Council of Medical Research, New Delhi. A written informed consent was obtained from the parents for publication of this research article and any accompanying images. A copy of the written consent is available for review by the Editor of this journal.”
3. We have also added comments about the data availability at the end of the main text.
4. We have incorporated both the page numbers and line numbers in the document.

Please find our revised manuscript for your further perusal
Looking forward to a favorable response.

Best regards

Faruq

Dr Mohammed Faruq, MBBS
Scientist
Genomics and Molecular Medicine,
CSIR-Institute of Genomics and Integrative Biology (CSIR-IGIB),
Mall Road, New Delhi, India.
Tel: 91-11-27666156; Fax: +91-11-27667471
Email: faruq.mohd@igib.res.in